Author's response to reviews

Title: A Rare coexistence of giant splenic hemangioma with diffuse liver hemangiomatosis in a patient with ovarian endometriosis.

Authors:

George GC Chatzoulis (surgchatz@yahoo.gr)
Andreas Kaltsas (akaltsas@hotmail.com)
Stauros Daliakopoulos (sdaliak@otenet.gr)
Osama Sallam (sallam@otenet.gr)
Maria Kaltsa (kaltsa.maria@gmail.com)
Kostas Chatzoulis (kxatzo@otenet.gr)

Version: 4 Date: 15 July 2007

Author's response to reviews:

424 General Military Hospital
Grigoriou Labraki 3(54638), THESSALONIKI,
GREECE.

Journal of Medical Case Reports
Thessaloniki 07/15/2007

Dear Sirs,

According to your instruction we revised our manuscript's figures and subsequent figures' legends. Written consent was obtained from the patient for publication of this case report. Thank you very much for your patience and kindness.

This manuscript has not been published or submitted for publication elsewhere. Thank you very much for your time.

Looking forward to hearing from you.

Best regards,

Dr George Chatzoulis and Colleagues